Author's response to reviews

Title: Genetic diagnosis of X-linked dominant hypophosphatemic rickets in a cohort study: Tubular reabsorption of phosphate and 1,25(OH)2D serum levels are associated with PHEX mutation type

Authors:

Marcos Morey (marcos.morey@usc.es)
Lidia Castro-Feijóo (lidia.castro.feijoo@usc.es)
Jesús Barreiro (jesus_barreiroconde@yahoo.es)
Paloma Cabanas (paloma.cabanas.rodriguez@sergas.es)
Manuel Pombo (manuelarturo.pombo@usc.es)
Marta Gil (marta.gil.calvo@sergas.es)
Ignacio Bernabeu (ignacio.bernabeu.moron@sergas.es)
José M Díaz-Grande (jose.miguel.diaz.grande@sergas.es)
Lourdes Rey-Cordo (m.carmen.lourdes.rey.cordo@sergas.es)
Gema Ariceta (M.GEMA.ARICETAIRAOLA@osakidetza.net)
Itxaso Rica (ITXASO.RICAECHEVARRIA@osakidetza.net)
José Nieto (jnieto@vhebron.net)
Ramón Vilalta (rvilalta@vhebron.net)
Loreto Martorell (lmartorell@hsjdbcn.org)
Jaime Vila-Cots (jvilacots@hsjdbcn.org)
Fernando Aleixandre (faleixandre@gmail.com)
Ana Fontalba (afontalba@humv.es)
Leandro Soriano-Guillén (leansor4@hotmail.com)
José M García-Sagredo (jgarcias.hrc@salud.madrid.org)
Sixto García-Miñaur (SIXTO719@telefonica.net)
Berta Rodríguez (berta.rodriguez.sanchez@sergas.es)
Saioa Juaristi (SAIOA.JUARISTIIRRURETA@osakidetza.net)
Carmen García-Pardos (CARMEN.GARCIA.PARDOS@osakidetza.net)
Antonio Martínez-Peinado (antonio.martinez.peinado.sspa@juntadeandalucia.es)
José M Millán (chema_millan@yahoo.es)
Ana Medeira (ana.medeira@hsm.min-saude.pt)
Oana Moldovan (moldovan.oana@gmail.com)
Ana Fernández (mfernandez161@yahoo.es)
Lourdes Loidi (lourdes.loidifernandez@sergas.es)

Version: 6 Date: 22 August 2011

Author's response to reviews:

Dear Editor,

I am enclosing the final revised version of the manuscript entitled “Genetic diagnosis of X-linked dominant hypophosphatemic rickets in a cohort study: Tubular reabsorption of phosphate and 1,25(OH)2D serum levels are associated
with PHEX mutation type” which we think is now ready for publication.
We have checked the manuscript carefully for any typing errors. In addition we
have made the formatting changes required:
- The author list in the manuscript is written exactly as it is in the submission
  system.
- Table 3 has been cited in the text.
- We have eliminated the shading in the tables as well as the vertical lines.
- References are cited in numerical order (Ref. 21 and 22 appear in table 1 and
  Ref. 24 and 25 in table 2. They do not appear in the text).
- The format of the Email addresses on the title page has been changed.
- The reference list contain up to 30 authors.
- We have cropped figure 3. Figures 1 and 2 can not be cropped.

With best regards,

Lourdes Loidi